Author's response to reviews

Title: Cross-sectional study of hepatitis B virus infection in rural communities, Central African Republic

Authors:

Narcisse P Komas (npkomas@yahoo.fr)
Ulrich Vickos (vickos.ulrich@gmx.com)
Judith M Hübschen (judith.huebschen@crp-sante.lu)
Aubin Béré (berostau@yahoo.fr)
Alexandre Manirakiza (amanirak@yahoo.fr)
Claude P Muller (Claude.Muller@crp-sante.lu)
Alain Le Faou (alainlefaou@univ-lorraine.fr)

Version: 5 Date: 10 June 2013

Author's response to reviews:

Dear Editor,

Thank you for your comments about our revised manuscript, MS: 1593046211922741. See enclosed our response to your comment:

Comment:
(1) Copyediting
"After reading through your manuscript, we feel that the quality of written English needs to be improved before the manuscript can be considered further. We advise you to seek the assistance of a fluent English speaking colleague, or to have a professional editing service correct your language. Please ensure that particular attention is paid to the abstract."

Response:
The critical reading and help in the revision of the quality of written English have been done by Professor Elisabeth Heseltine. She paid particular attention to the abstract.
The manuscript is now conform to the journal style.

Dr Narcisse P Komas